Author’s response to reviews

Title: THE EAGLE JUGULAR SYNDROME

Authors:
Paolo Zamboni (zambo@unife.it)
Alba Scerrati (a.scerrati@gmail.com)
Erica Menegatti (mngerce@unife.it)
Roberto Galeotti (roberto.galeotti@unife.it)
Marcello Lapparelli (m.lapparelli@ospfe.it)
Luca Traina (l.traina@ospfe.it)
Mirko Tessari (mirko.tessari@gmail.com)
Andrea Ciorba (andrea.ciorba@unife.it)
Pasquale De Bonis (dbnpql@unife.it)
Stefano Pelucchi (stefano.pelucchi@unife.it)

Version: 2 Date: 09 Dec 2019

Author’s response to reviews:

Editor Comments:

Q1. It is noticed that your figures and supplementary files are not visible due to an error. Upon revision please ensure all figures and supplementary files are uploaded correctly and are visible.

A1. We re-uploaded the figures and supplementary files and now they are visible.

Q2. Please provide a “List of abbreviations” subsection after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.
A2. We provided the “List of abbreviations” after the conclusions (page 12) and defined them along in the text.

Q3. Please add a heading "Declarations" after your List of abbreviations and before the References section. Under this heading be sure to add the following sub-sections:
a. Ethics approval and consent to participate
b. Consent for publication
c. Availability of data and materials
d. Competing interests
e. Funding
f. Authors’ contributions
g. Acknowledgements
h. Authors' information (optional)

A.3 We added the “declarations” section in the text (page 13) and defined the requested subsections.

Q4. As indicated in our submission guidelines, please include the following sections in your manuscript: Background, Methods, Results, Discussion, Conclusions.
A4. We included the requested sections in the paper.

Q5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
A5. We uploaded the manuscript as requested by the editor.